Author's response to reviews

Title: Analysis of mutations in the E6 oncogene of Human Papillomavirus 16 in cervical cancer isolates from Moroccan women

Authors:

Zineb Qmichou (zineb.qmichou@gmail.com)
Meriem Khyatti (meriem.khyatti@pasteur.ma)
Mohamed Berraho (maberraho@yahoo.fr)
Moulay Mustapha Ennaji (m.ennaji@yahoo.fr)
Laila Benbacer (lbenbacer@yahoo.com)
Chakib Nejjari (cnejjari2000@yahoo.fr)
Noureddine Benjaafar (benjaafar@menara.ma)
Abdellatif Benider (beniderabdell@yahoo.fr)
Mohammed Attaleb (attaleb_mohammed@yahoo.fr)
Mohammed El Mzibri (mzibri@yahoo.com)

Version: 3 Date: 13 May 2013

Author's response to reviews:

Dear Professor,

I would like to submit to BMC Infectious Disease Journal a manuscript untitled “Analysis of mutations in the E6 oncogene of Human Papillomavirus 16 in cervical cancer isolates from Moroccan women”. This article has been written by: Zineb Qmichou, Meriem Khyatti, Mohamed Berraho, My Mustapha Ennaji, Laila Benbacer, Chakib Nejjari, Noureddine Benjaafar, Abdellatif Benider, Mohammed Attaleb and Mohammed El Mzibri.

As requested by the editorial office, we have modified the main text to include the name of the ethics committee that has approved the study.

Best regards